Author's response to reviews

Title: Can patient decision aids help people make good decisions about participating in clinical trials? A study protocol

Authors:

Jamie C Brehaut (jbrehaut@ohri.ca)
Alison Lott (alott@ohri.ca)
Dean A Fergusson (daferfusson@ohri.ca)
Kaveh G Shojania (kshojania@ohri.ca)
Jonathan Kimmelman (jonathan.kimmelman@mcgill.ca)
Raphael Saginur (rsaginur@ottawahospital.on.ca)

Version: 8 Date: 11 July 2008

Author's response to reviews:

Small typo changes